\documentclass[letterpaper,12pt]{article}

\usepackage{amsfonts,amssymb,amsmath,amsthm,txfonts,pxfonts,mathrsfs,fullpage,enumitem}
\usepackage{xcolor}

\pagestyle{empty}
\begin{document} 

\begin{center}\textbf{Problem Set 9 \hspace{.7in} Foundations \hspace{.7in} Due 4/27/16} \end{center}

\noindent 20 points possible

\begin{enumerate}
\item Pick at least 12 problems that you recommend be included on our final exam, including at least 2 from each chapter in the text.  Give good reasons why each should be included.

\item List at least 3 areas where you currently feel unprepared for the final exam.

\end{enumerate}

\end{document}